Author’s response to reviews

Title: Swallowing transit times and valleculae residue in stable chronic obstructive pulmonary disease

Authors:
Rosane D Chaves (rosanechaves@uol.com.br)
Fernanda C Sassi (fsassi@usp.br)
Laura D Mangilli (lauramangilli@usp.br)
Shri K Jayanthi (skjayanthi@yahoo.com.br)
Bruno Zilberstein (brunozilb@uol.com.br)
Claudia RF Andrade (clauan@usp.br)

Version: 2 Date: 12 December 2013

Author’s response to reviews:

São Paulo, 12th December 2013
Catia Cornacchia
Executive Editor
BMC Pulmonary Medicine
Ref.: submission of revised manuscript

Enclosed is the revised version of our paper titled “Swallowing transit times and valleculae residue in stable chronic obstructive pulmonary disease”, by Rosane de Deus Chaves, Fernanda Chiarion Sassi, Laura Davison Mangilli, Shri Krishna Jayanthi, Bruno Zilberstein and Claudia Regina Furquim de Andrade.

As requested by the Editorial Office of BMC Pulmonary Medicine, we have included the following sections in our paper: Acknowledgements and Funding.

Thank you for your attention to our paper.

Best regards,

Prof. Claudia Regina Furquim de Andrade (on behalf of all authors)